Author's response to reviews

Title: A Qualitative Exploration of Malaysian Cancer Patients' Perspectives on Cancer and its treatment

Authors:

Maryam Farooqui (maryamfaroocui.uitm@gmail.com)
Mohamed Azmi Hassali (azmihassali@gmail.com)
Aishah Knight Shatar (Aknight@amdi.usm.edu.my)
Asrul Akmal Shafie (aakmal@usm.my)
Tan Boon Seang (bseang@yahoo.com.sg)
Muhammad Aslam Farooqui (farooqui@acms.edu.my)

Version: 6 Date: 30 June 2011

Author's response to reviews: see over
The Editor  
BMC-series Journals  
BioMed Central  
London  

Thank you very much for your comments on my manuscript MS: 5481827975157456. Below are my responses to the requests made by the editorial board:

Comments: Author list: the list of authors in the manuscript should be written exactly as they are in the submission system, both in style and order. The preferred style is 'First name Initial Last name' (e.g. Joe F Bloggs)
Response: Modified as suggested

Comments: References: Please make sure references are cited in numerical order. We notice reference 20 is not cited within the main text. Please make sure references continue to be cited in numerical order once this reference is cited.
Response: Modified as suggested

Comments: Structure: Please don't make 'Limitations' as a heading. Change to a subheading.
Response: Modified as suggested

Comments: Typography: Please take this opportunity to check your manuscript for any typographical errors and to make any final corrections or revisions. This is the final proofing stage for your manuscript, and you will not be able to make any changes after acceptance.
Response: The manuscript was checked few times for typing errors. I hope the latest version is free from mistakes.

Thank you very much, I look forward for your kind response

Maryam Farooqui  
Faculty of Pharmacy  
Universiti Teknologi MARA (UiTM), Pulau Pinang  
13500, Seberang Perai, Pulau Pinang  
Malaysia